Author's response to reviews

Title: Orthodontic treatment of a severe anterior open bite and an alveolar bone defect caused by incisor trauma: A case report

Authors:

Feiou Lin (499249858@qq.com)
Hao Sun (sunhao19891218@gmail.com)
Linjie Yao (1303171089@qq.com)
Qiushuo Chen (42014335@qq.com)
Zhenyu Ni (nzy1993@163.com)

Version: 3
Date: 12 August 2014

Author's response to reviews: see over
Dear Editor,

I along with my co-authors, would like to re-submit the attached manuscript entitled “Orthodontic treatment of a severe anterior open bite and an alveolar bone defect caused by incisor trauma: A case report” as an original article.

The manuscript has been carefully rechecked and appropriate changes have been made in accordance with the editor’s comments and these changes have been highlighted. We hope that the revised manuscript is now suitable for publication in your journal.

I look forward to your reply.

Sincerely,
Zhenyu Ni
Orthodontic Department, School of Stomatology, Wenzhou Medical University No. 113 West Xueyuan Road, Wenzhou, Zhejiang, China
Phone and Fax: +86-577-88063008;
E-mail: nzy1993@163.com